Author's response to reviews

Title: Locked patellar dislocation: a case report

Authors:

Frederick Michels (frederick_michels@hotmail.com)
Nicole Pouliart (NicolePouliart@uzbrussel.be)

Version: 2 Date: 10 April 2008

Author's response to reviews:

Dear editor,

Thank you for the useful comments about our abstract. As you asked we removed the abbreviations in the abstract, we included information on the patient's ethnic origin, we restructured the Introduction section as requested.

Hopefully this modifications will allow further review.

Sincerely yours,

Corresponding author,

Frederick Michels, frederick_michels@hotmail.com